Author's response to reviews

Title: Severe and isolated headache associated with severe hypertension as unique symptom of posterior reversible encephalopathy syndrome

Authors:

Gregorio P Milani MD (yoyobiancorosso@hotmail.com)
Alberto Edefonti MD (alberto.edefonti@policlinico.mi.it)
Giacomo Tardini MD (giacomo.tardini@gmail.com)
Elisa Arturi MD (elisa.arturi@gmail.com)
Claudia M Cinnante MD (claudia.cinnante@policlinic.mi.it)
Alberto Cappellari MD (alberto.cappellari@policlinico.mi.it)
Emanuela A Laicini MD (emanuela.laicini@policlinico.mi.it)
Ernesto Leva MD (ernesto.leva@policlinico.mi.it)
Carlo Agostoni MD (agostoc2@tin.it)
Emilio F Fossali MD (emilio.fossali@unimi.it)

Version: 5  Date: 5 July 2014

Author's response to reviews:

Manuscript (Case report): “Severe and isolated headache associated with severe hypertension as unique symptom of posterior reversible encephalopathy syndrome” (Gregorio Paolo Milani, Alberto Edefonti, Giacomo Tardini, Elisa Arturi, Claudia Maria Cinnante, Emanuela Anna Laicini, Ernesto Leva, Alberto Maria Cappellari, Carlo Agostoni, Emilio Filippo Fossali)

Dear Dr. Catherine Olino:

Thank you very much for your recent letter. We were pleased to know that the concerns raised by reviewers have been rightly addressed and have improved the quality of the manuscript. In this revision we have corrected some typographical errors as requested by Reviewer # 1. We edit the language as requested by editorial boarding to improve the English style. Finally, we have updated the references with a recent publication about PRES written by some of us (Reference number 10).

Therefore we hope that with these corrections our manuscript is acceptable for publication in BMC Pediatrics. Please do not hesitate to contact me again if you feel that further corrections are required.

Best regards.

Gregorio Milani, 5th of July, 2014